Author's response to reviews

Title: Examining intra-rater and inter-rater response agreement: A medical chart abstraction study of a community-based asthma care program

Authors:

Teresa To (teresa.to@sickkids.ca)
Eileen Estrabillo (eileen.estabillo@utoronto.ca)
Chengning Wang (chengning.wang@sickkids.ca)
Lisa Cicutto (lisa.cicutto@utoronto.ca)

Version: 4 Date: 2 May 2008

Author's response to reviews: see over
Dear Dr. Kowalczuk,

Thank you for reviewing our manuscript entitled “Examining Intra-Rater and Inter-Rater Response Agreement: A Medical Chart Abstraction Study of a Community-Based Asthma Care Program”. We have addressed and incorporated all of the new comments from the editors in this revised version and would like to resubmit it for consideration for publication in BMC-Medical Research Methodology.

The following are a listing of our responses to the editors’ comments.

1) “Author’s contributions” → “Authors’ contributions”  [Changes are made on page 15]

2) “Figures” → “Figure legends”  [Changes are made on page 19]

3) Figures should be cropped as closely as possible.  [Changes are made to both figures]

We hope that our responses to the editors’ comments and resulting revisions to the manuscript will be to the satisfaction of the Editors. Please do not hesitate to contact me should you have any questions related to this manuscript.

On behalf of the co-authors I would like to thank you again for your consideration.

Sincerely yours,

Teresa To, BSc, MSc, PhD
Senior Scientist and Head, Child Health Evaluative Sciences, Hospital for Sick Children
Associate Professor, Department of Public Health Sciences, U of Toronto
Telephone:  (416)813-8498
Fax:  (416)813-5979
Email:  teresa.to@sickkids.ca